Author's response to reviews

Title: Study of Sedative activity of different extracts of Kaempferia galanga in Swiss albino mice

Authors:

Mohammad S Ali (shawkat.ali@bracu.ac.bd)
Pritesh R Dash (pritesh@bracu.ac.bd)
Mahmuda Nasrin (munninasrin@yahoo.com)

Version: 2 Date: 11 February 2015

Author's response to reviews: see over
To
The Editor in Chief
BMC Complementary and Alternative Medicine
BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Sub: Submission of Revised Manuscript (MS: 8512465413538524).

Dear Sir,

Thank you very much for your kind review of the manuscript entitled “Study of Sedative activity of different extracts of Kaempferia galanga in Swiss albino mice”. MS: 8512465413538524. The manuscript is revised as per reviewer suggestion and submitting to you for publication. However following changes are made in the revised manuscript:

1. Title is rewritten.
2. In the abstract, background is rewritten and description for the extracts removed from this part and added to it in methods part. In the conclusion the effects of different extracts are added.
3. Kew words are rewritten.
4. In the title of Drugs and Treatment, extracts are reconstituted in water.
5. Previously for neuropharmacological study, dose of Diazepam was 1mg/kg, but now for sedative activity study dose of Diazepam 2mg/kg is added.
6. Background and Discussion parts are revised.
7. Conclusion is rewritten and study of the results is summarized.
8. Spelling, grammar, punctuation error (s) and abstract, background, discussion and conclusion parts are checked.
9. All authors e-mail address is provided.

All changes in the revised manuscript are indicated by red mark.

Thanking you.

Sincerely Yours

Dr. Mohammad Shawkat Ali
Chairperson and Professor
Department of Pharmacy, BRAC University
41, Pacific Tower, Mohakhali, Dhaka, Bangladesh
Cell: 01712047064
E-mail: shawkat.ali@bracu.ac.bd